Author's response to reviews

Title: The co-administration of quetiapine or placebo to cognitive-behavior therapy in treatment refractory depression; a preliminary trial

Authors:

Yves JA Chaput (yveschaput@bellnet.ca)
Annick Magnan (tobedetermined@bellnet.ca)
Alain Gendron (alaingendron@astrazeneca.ca)

Version: 5 Date: 4 August 2008

Author's response to reviews: see over
To the Editorial Staff,

BMC – Psychiatry,

MS: 8312126711678731

Montreal, August 4th, 2008.

Thank you for reviewing and accepting my manuscript entitled ‘The co-administration of quetiapine or placebo to cognitive-behavior therapy in treatment refractory depression; a preliminary trial’ that I re-submitted for your consideration in BMC-Psychiatry a few weeks ago.

I have not modified the text other than slight formatting changes as well as correcting several typos. I have also looked at the references to format them in accordance with Biomed Central requirements.

Figure 1, the flowchart, has been altered a bit esthetically also.

In the hope that this revision will meet with your satisfaction,

Yves Chaput, MD, FRCP(C), PhD